Author's response to reviews

Title: Clinical approaches to treating papillary squamous cell carcinoma of the uterine cervix

Authors:

Michikazu Nagura (mnagura@kuhp.kyoto-u.ac.jp)
Masafumi Koshiyama (koshiyamam@nifty.com)
Noriomi Matsumura (noriomi@kuhp.kyoto-u.ac.jp)
Aki Kido (akikido@kuhp.kyoto-u.ac.jp)
Tsukasa Baba (babatsu@kuhp.kyoto-u.ac.jp)
Kaoru Abiko (kaoruvc@kuhp.kyoto-u.ac.jp)
Junzo Hamanishi (jnkhmns@kuhp.kyoto-u.ac.jp)
Ken Yamaguchi (soulken@kuhp.kyoto-u.ac.jp)
Yoshiki Mikami (mika@kuhp.kyoto-u.ac.jp)
Ikuo Konishi (konishi@kuhp.kyoto-u.ac.jp)

Version: 5 Date: 15 September 2014

Author's response to reviews: see over
Editorial Comments

We corrected the sentences with underline in manuscript.

1. Human Ethics

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Answer>>> We add the following sentence: This study was approved by the ethics committees of Kyoto University Hospital (No. G288), and all patients provided written informed consent prior to study entry, on 94 line

2. Acknowledgements

By way of a section ?Acknowledgements?, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript: and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest
wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.' Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

Answer>>> We add the following sentence:

**Acknowledgements**

We would like to thank Kosaka K, Shizuoka General Hospital and Kataoka M, Kyoto University Hospital for their support and provision of the email address list. on line 330.

3. Copy Edit

Please note that all submissions to BMC Cancer must comply with our editorial policies. In particular, please ensure that your manuscript contains any required statements regarding ethical approval and consent from patients/ participants; adheres to any relevant standards of reporting with a completed checklist where appropriate; and contains an Availability of Supporting Data section explaining how the supporting data is available to researchers, including the repository and accession number where appropriate. Manuscripts which do not adhere to our editorial policies may experience delays or be rejected. Further information about our editorial policies can be found at the following links:

Ethical approval and consent:
http://www.biomedcentral.com/about/editorialpolicies#Ethics

Standards of reporting:
http://www.biomedcentral.com/about/editorialpolicies#StandardsofReporting

Data availability:
http://www.biomedcentral.com/about/editorialpolicies#DataandMaterialRelease

Please ensure that all changes to the manuscript are indicated in the text by highlighting or using track changes.

Answer>>> We add the following sentence:  In this connection, the
The manuscript was observational and diagnostic accuracy study, on line 88.

We add the following sentence: The primary surgeries were performed to obtain the final diagnosis after having got informed consents from the patients, on line 100.

We add the following sentence: For all manuscripts that included details or images relating participants, written informed consent for the publication of these were obtained from the participants, on line 208.

4. Please can you clarify whether you received consent to publish the images and clinical details in this manuscript from the patients in question. If you did obtain consent to publish, please clarify this within your manuscript.

We add the following sentence: For all manuscripts that included details or images relating participants, written informed consent for the publication of these were obtained from the participants, on line 208.